Author’s response to reviews

Title: Dose patterns in commercially insured subjects chronically exposed to opioids: a large cohort study in the United States

Authors:

Maria Soledad Cepeda (scepeda@its.jnj.com)
Mila Etropolski (metropol@its.jnj.com)
Rachel Weinstein (rweinst1@its.jnj.com)
Daniel Fife (dfife@its.jnj.com)
Raymond Boston (drrayboston@yahoo.com)
Amy Matcho (amatcho@its.jnj.com)

Version: 10 Date: 15 June 2010

Author’s response to reviews: see over
Major revisions
Author list: the list of authors in the manuscript should be written exactly as it is in the submission system, both in style and order. The preferred style is ?First name Initial Last name? (e.g. Joe F Bloggs).

Response:
I spelled out the name of the first author. However, this author uses the middle name, so would it be possible spell out the middle name as well?

Abstract: please ensure that the Abstract is present in the manuscript and matches exactly what is in the submission system.

Response:
The abstract was added

Minor revisions
Email address: For the authors’ email address list, please use the following format as an example:

Response:
The e-mail addresses now have the required format

Highlighting: Please remove.

Response:
It was removed

Authors' contributions: Please format into a paragraph.

Response: Done

References: the reference list should contain up to 30 authors' names for each citation. The term ?et al.? should not be used if there are fewer than 30 authors.

Response: All the references have all the authors listed

Tables: Please move to after reference list.

Response: Done

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Response: We have checked the manuscript for errors and we have corrected them